\begin{table}%t7
      \caption{Measured values for reddening of V5114 Sgr.}
         \label{ReddTabl}
         %\centering
\small
       \begin{tabular}{l l l}
            \hline
            \hline
            method & reference &  $E_{B-V}$      \\
            \hline
	$(B-V)_{\rm max}$	& van den  Bergh  \&  Younger (\cite{vdBY}) &  0.43 \\
	$(B-V)_{t_2}$	& van den  Bergh \&  Younger  (\cite{vdBY}) &  0.40 \\
	\ion{K}{i} $\lambda$7699& Munari  \&  Zwitter (\cite{Munari})   &  0.45\\
	\ion{O}{i} emission line ratios	& Rudy et~al. (\cite{Rudy91})	&  1.23 \\
	Dust maps	&	Schlegel et~al. (\cite{Schlegel}) &  0.58 \\
	\ion{He}{ii}$\lambda\lambda$10124, 4686 & Williams
	(\cite{Williams94})  &  0.57 \\
	\ion{H}{I}$\lambda\lambda$10049, 4861 & Williams
	(\cite{Williams94}) & 0.65 \\
	\ion{He}{i}$\lambda\lambda$10830, 4471 &  Osterbrock
	\cite{Osterbrock}  & 0.9 \\
	\ion{He}{i}$\lambda\lambda$5876, 4471 &  Robbins
	(\cite{robbins}) & 0 \\
            \hline
         \end{tabular}
   \end{table}